

Working Paper Series **J5**

Demand for Foreign Exchange Derivatives in Brazil: Hedge or Speculation? Fernando N. de Oliveira and Walter Novaes

December, 2007

Working Paper Series	Brasília	n. 152	Dec	2007	P. 1-47
U					

ISSN 1518-3548 CGC 00.038.166/0001-05

Working Paper Series

Edited by Research Department (Depep) - E-mail: workingpaper@bcb.gov.br

Editor: Benjamin Miranda Tabak – E-mail: benjamin.tabak@bcb.gov.br Editorial Assistent: Jane Sofia Moita – E-mail: jane.sofia@bcb.gov.br Head of Research Department: Carlos Hamilton Vasconcelos Araújo – E-mail: carlos.araujo@bcb.gov.br

The Banco Central do Brasil Working Papers are all evaluated in double blind referee process.

Reproduction is permitted only if source is stated as follows: Working Paper n. 152.

Authorized by Mário Mesquita, Deputy Governor for Economic Policy.

General Control of Publications

Banco Central do Brasil Secre/Surel/Dimep SBS – Quadra 3 – Bloco B – Edifício-Sede – 1° andar Caixa Postal 8.670 70074-900 Brasília – DF – Brazil Phones: (5561) 3414-3710 and 3414-3567 Fax: (5561) 3414-3626 E-mail: editor@bcb.gov.br

The views expressed in this work are those of the authors and do not necessarily reflect those of the Banco Central or its members.

Although these Working Papers often represent preliminary work, citation of source is required when used or reproduced.

As opiniões expressas neste trabalho são exclusivamente do(s) autor(es) e não refletem, necessariamente, a visão do Banco Central do Brasil.

Ainda que este artigo represente trabalho preliminar, citação da fonte é requerida mesmo quando reproduzido parcialmente.

Consumer Complaints and Public Enquiries Center

Address:	Secre/Surel/Diate
	Edifício-Sede – 2º subsolo
	SBS – Quadra 3 – Zona Central
	70074-900 Brasília – DF – Brazil
Fax:	(5561) 3414-2553
Internet:	http://www.bcb.gov.br/?english

Demand for Foreign Exchange Derivatives in Brazil: Hedge or Speculation?

Fernando N. de Oliveira^{*} Walter Novaes^{**}

The Working Papers should not be reported as representing the views of the Banco Central do Brasil. The views expressed in the papers are those of the author(s) and do not necessarily reflect those of the Banco Central do Brasil.

Abstract

This paper examines empirically the demand of foreign exchange derivatives by Brazilian corporations. We build an original database of 25,457 contracts of foreign exchange swaps between firms and financial institutions open at the end of 2002. From these contracts we identify 53 corporations that hedge in the foreign exchange derivatives market and 40 corporations that speculate. The data show that the existence of external debt and the size of the company affect positively the probability of hedging, whereas revenues from exports affect positively the probability of speculation. These results suggest that during periods of great volatility of the exchange rate – such as in 2002 – the corporations' demand for foreign exchange derivatives is strongly related to speculative motives.

Keywords: foreign exchange swaps, hedge, speculation, corporations **JEL Classification:** G13, G32, G38

^{*} Research Department, Banco Central do Brasil. E-mail: fernando.nascimento@bcb.gov.br

^{**} Pontifícia Universidade Católica do Rio de Janeiro – PUC/RJ.

1. Introduction

Géczy, Minton e Schrand (1997) show that 41.4% of North American companies pertaining to the Fortune 500 group use foreign exchange derivatives in 1990. These financial instruments impose gains and losses to these companies according to the variation of the nominal exchange rate. What makes these North American firms add financial risk to their operations?

The risk of an isolated asset does not necessarily increase the risk of a portfolio of assets. On the contrary, assets with a significant level of risk such as foreign exchange derivatives can induce cash flow fluctuations that nullify the risk of individual assets essential to a company's operations. We refer to this risk management strategy as a *hedge*. A common explanation for the frequent use of foreign exchange derivatives in the USA is, therefore, that firms seek to diminish the volatility of their cash flows.

In fact, Géczy, Minton e Schrand (1997) show that firms with a high cost of financial stress and with high cash flow volatility are more likely to use foreign exchange derivatives, which suggests that these firms are trying to protect themselves from changes in the exchange rate, that can induce losses capable of taking them to the point of financial stress. Géczy, Minton e Schrand (1997), however, do not explicitly show that the use of foreign exchange derivatives diminishes a firm's foreign exchange risk. It is possible that, for example, this use is in response to opportunities for speculative gain derived from private sources (see Merton (1987)). In this respect, foreign exchange derivatives can increase a firm's foreign exchange risk, reflecting a demand for speculative measures.

This paper builds an original database of 74,567 contracts of foreign exchange swaps written between corporations and financial institutions from January 1999 to December 2002. Of these 25,457 are open at the end of 2002. This database allows us to identify the use of foreign exchange derivatives in order to hedge and in order to speculate in 2002. This year seems appropriate due to an enormous depreciation of the foreign exchange rate during the year (from the start to the end of the year there was a 60% depreciation). In a situation like this the incentives for speculation or hedge are of course much more evident.

By identifying these two demands for derivatives, we show that the existence of debt in a foreign currency is the principal determinant of the demand to hedge for 53

Brazilian corporations with open contracts of foreign exchange derivatives in 2002, whereas gross revenues from exports is the principal determinant of speculation for another 40 Brazilian corporations. Thus, the data suggest that during periods of great exchange rate volatility – such as in 2002 – the firms' demand for foreign exchange derivatives is strongly related to speculative motives.

We compare our results of the year 2002 with the results of hedge and speculation of the years from 1999 to 2001. In these years, the foreign exchange rate was much less volatile and average depreciation much less pronounced than in 2002. We observe that the year 2002 is atypical year as far as speculation is concerned. Not only the number of firms that speculated during these years is much less than the number of firms that speculated in 2002 but also exports are not anymore the driving force of speculation.

The data used in this study is obtained from confidential information of Banco Central do Brasil. Additional available data from the two institutions that register the total volume of operations of foreign exchange derivates in Brazil – Brazilian Mercantile & Futures Exchange (BM&F) and CETIP Custody and Settlement – show that, between 1999 and 2002, the contracts for exchange of interest denominated in reals for dollar-denominated interest (foreign exchange swaps) are by far the most important instrument of foreign exchange derivatives used by companies with operations in Brazil. Based on this information, the Banco Central do Brasil solicited 50 authorized financial institutions that operate in the foreign exchange derivatives to inform the name of all the companies with which they signed contracts of foreign exchange swap from January 1999 to December 2002 as well as the notional amount, currency and maturity of these contracts.

In response to these information solicited by the Central Bank, 43 financial institutions described details of 74,567 contracts of foreign exchange swaps. Of these 25,457 are still outstanding at the end of 2002, the year in which the exchange rate increased from R\$2.31 per dollar in January to R\$3.50 per dollar in December – a depreciation of 60.0%, which in part was due to uncertainties regarding the elections. The year 2002 seems to be a good candidate, therefore, to capture uses of foreign exchange derivatives for speculation as well as for hedging.

Our results show that 93 public owned firms had currency swaps positions open at the end of 2002. Of these 93 corporations, 40 held speculative positions for swaps that increase their foreign exchange risk. We classify speculation in two different types: reverse speculation and neutral speculation. Reverse speculation occurs when a firm holds open positions in swaps contrary to its operational currency exposure (difference between export revenues in dollars and the sum of imports expenses in dollars and debt in dollars). That is, companies that, due to operational currency exposure, lose (gain) with an appreciation (depreciation) of the exchange rate are long (short) in swaps, magnifying these monetary losses. Of 40 firms with speculative positions, 16 did reverse speculation. We define neutral speculation when the firm does not have operational currency exposure but still has open foreign exchange swaps positions. Of the 40 firms that speculated, 24 did neutral speculation.

On the other hand, of the 93 companies with contracts for open currency swaps 53 intend to diminish their currency exposure. Among these companies that hedge, 35,84% are concentrated in the public service sector and all have dollar-denominated debt.

The main contribution of this paper comes from the use of a unique database of foreign exchange derivatives¹. This database makes it possible to get a much better understanding of which firms speculate and those that hedge the foreign exchange exposure. This is very rare in the literature that studies the demand of foreign exchange derivatives by firms. This literature normally looks at off balance sheet information of firms².

The rest of the article is organized as follows: Section 2 describes the data and presents the results of the univariate analysis. Section 3 describes tests of multivariate analysis and presents the results. Section 4 analyzes the robustness of the results, and finally section 5 presents the main conclusions of the paper.

¹ There are other papers that study the demand of foreign exchange derivatives in Brazil, like for example Shiozer and Saito (2006). The authors use off balance sheet information of firms. This paper studies the reasons firms hedge with foreign exchange derivatives. Our paper differs because our main objective is to compare firms that hedge with firms that speculate with foreign exchange derivatives using a database that was built with swap contracts of foreign exchange.

 $^{^2}$ See Tirole (2006) for a discussion of the problems related to off balance sheet information of corporations.

2. The Data

Our primary data source is a unique database, composed of 74,567 contracts of currency swaps signed between 1999 and 2002 between 43 financial institutions and non-financial corporations in Brazil. These contracts correspond to nearly 98% of the total volume of currency swaps transacted in 2002^3 .

In order to understand the importance of our database, we need to explain briefly the structure of the Brazilian foreign exchange derivatives market. There are various types of foreign exchange derivatives used by firms and financial institutions: public bonds indexed to the dollar, operations with foreign exchange futures, options and forwards. In the currency swap contracts, the investor in the long position trades interest in reals for dollar-denominated interest; this implies gains (losses) with a depreciation (appreciation) of the exchange rate.

The initial demand for public bonds indexed to the dollar is made by the financial institutions and is registered in the System of Liquidation and Custody of Federal Public Securities (SELIC). The other derivatives are registered at Brazilian Mercantile and Futures Exchange (BM&F) or at the CETIP Custody and Settlement.

The main contracts of registered firms at BM&F are future contracts and dollar options. According to available data at BM&F, dollar-denominated future contracts are only liquid for maturities within 20 days and their open total daily stock is almost always less than 3% of total open stock of currency swaps between firms and financial institutions registered at CETIP. Dollar options are even less liquid and present daily stock levels even lower than those of dollar-denominated future contracts.

The fact that currency swaps are the main instrument of foreign exchange derivatives used by corporations can be explained in part by simply observing the data. The data show that a great number of corporations that use foreign exchange derivatives have debt in a foreign currency. In general, this debt has middle to long-term maturities, with disbursement of interests done irregularly. Futures contracts, options or forward contracts of foreign exchange with long maturities are, in general, not liquid or

³ The Central Bank initially solicited information from 50 financial institutions. Some of these institutions were purchased by others of the group of 50, between 1999 and 2002. The purchasing financial institutions became responsible for the information regarding the contracts for currency swaps of those

inexistent, arising from this the need for corporations to demand counter operations, such as swaps, that better reflect the cash flow of their external obligations⁴.

The total volume of transacted currency swaps between firms and financial institutions is quite superior to the volume negotiated among non-financial firms. This fact is not peculiar to Brazil. Mian (1996) shows that, in the majority of countries, non-financial corporations seek financial institutions as the other party in their derivatives operations. Available data of the CETIP show that, between 1999 and 2002, the daily stock of currency swaps among corporations is on average 3% of the daily stock of currency swaps done between financial institutions and corporations. Among the currency swaps, those for which the US dollar is one of the objects of operation represent more than ninety-five percent of the total volume negotiated⁵. Therefore, the Central Bank's database of US dollar-based currency swaps is fairly representative of the demand for foreign currency derivatives of Brazilian companies.

The empirical analysis will have as its focus the corporations that have open positions in currency swaps at the end of 2002. We combine these corporations with the following control group: all of the non-financial corporations that do not pertain to the same economic group and have some form of exchange rate exposure⁶. We consider a firm having exchange rate exposure if it has debt in foreign exchange, or exports or imports or is a part of a sector of the economy that has foreign exchange exposure. This group consists of 250 corporations that together with those that have open positions in currency swaps at the end of 2002, 93 corporations, comprise a total of 343 corporations⁷.

Table 1 shows the financial characteristics of the corporations that have open positions and of those that do not. The corporations with open positions are larger, show more debt in US dollars, have greater ratios of external revenue to gross revenue and

institutions that were purchased. This explains why the number of institutions that responded the initial request was 43 and not 50.

⁴ Other possible explanations are: the swaps do not need collateral (which is required by BM&F); they do not suffer daily adjusts and they also do not require an initial payment.

⁵ A great majority of these contracts have maturities of less than two years and on the other end of the contract are pre and post-fixed interest rates.

⁶ The selection has as base financial statements of the exercise of 2002, which became public and available at the Comissão de Valores Mobiliários (CVM). We choose those that furnished all the necessary accounting information.

⁷ Close to 90% of businesses in the sample are among the 1000 largest Brazilian companies in terms of net revenue in the year 2002, according to the annual Valor 1000 of August 2003.

have more executives participating in the profits. In all of these cases, the hypothesis of average equality is rejected at the 5% significance level.

Of the 93 corporations in our sample, 7 (7.13%) are multinationals. These corporations probably have a natural demand to hedge in order to protect the investment of their shareholders in the country of origin from fluctuations of the nominal exchange rate. By taking the multinationals out of the sample, the only difference is that the corporations with open positions in currency swaps in 2002 become smaller than those corporations without open positions.

Next, we define which of these corporations hedge or which speculate. The corporations that hedge are those for whom the product between the open net position in currency swaps and what we call operational currency exposure (the difference between export revenues in dollars and the sum of the imports expenses and debt in dollars) is less than zero. This product can be negative in two cases: when a corporation has an operational currency exposure greater than zero and tries to protect itself from a currency appreciation taking short positions in dollars; and when it has an operational currency exposure less than zero and tries to protect itself from a currency depreciation taking long positions in dollars.

The corporations that speculate are classified in two groups. We define reverse speculation when the product of the operational currency exposure and the net open position of the currency swap is greater than zero. This case includes the corporations whose value would diminish due to currency depreciation, even when they opt for short positions in currency swap contracts, and those whose value would increase with a currency appreciation and would nonetheless remain in long positions in currency swaps.

If a corporation does not possess operational currency exposure, it speculates if it holds an open long or short position in the currency swap. We call this neutral speculation. This latter type of speculative position is less reliable given that the company, in spite of apparently not having operational currency exposure, can have input or sales of products whose prices have a direct or maybe indirect relationship with the foreign exchange. Later on in our analyses we will control for this possibility.

9

In order to find the open net position of the firm in currency swaps, we transform all the values in reals to dollars, using the exchange rate of the date the operation. Next, we verify all of the open operations at the end of 2002 (long and short) and we find the net position of each corporation as the difference between the total volume in dollars of the long positions and the total volume in dollars of short operations⁸. Table 2 presents a classification by sectors of the corporations that hedge or that speculate, and shows the form they choose (long or short positions), in addition to information about the operational currency exposure of each of the different sectors.

Panel A of Table 2 shows that the number of corporations that speculate is a little less than the number of corporations that hedge. Among these, the number of neutral firms that speculate is greater than the number of those that speculate in reverse positions. All of the corporations that speculate hold long positions. This suggests expectations of currency depreciation. This occurs even among predominantly export sectors, such as food products and beverages. In terms of hedge, we can see that all corporations are long in dollars. Corporations that hedge are primarily from the concessionary of public service sector of the economy. This is the sector that in the aggregate shows the most significant debt in foreign exchange relative to its assets.

There are some sectors in which the average of the ratio between the values of open net positions in currency swaps and net worth is relevant. This occurs in both the case of the hedge (8.0% in the case of Electro/Electronic) and in the case of reverse speculation (4.0 % for the food and beverage sector). Given that the firms of these sectors are long in dollars, this fact reveals, once again, the expectation on the part of the firms of substantial currency depreciation during 2002.

Panel B of Table 2 shows that debt in dollars is ubiquitous among corporations that hedge. All of them have debt in dollars⁹. In addition, panel C of Table 2 shows that firms that export predominate in the case of reverse speculation. Some of these firms also show some imports expenses and debt in dollars, but in volumes inferior to those of their exports revenues. Finally, Panel D of Table 2 shows that 84.94% of the

⁸ We also tried the average value of the exchange rate of the month to transform the notional value of the contracts and the results did not change.

⁹ We also looked at "Adiantamento de Contratos de Câmbio", ACC. These are loan contracts that exporters write with banks in which the collateral is future exports. We can consider them as a form of foreign exchange debt contracts. Very few exporters in our database have these contracts, though. Our results did not change by considering them foreign exchange debt as well.

corporations that speculate or that hedge are private domestic firms, whereas only 7.93% are state-owned and 7.13% are multinationals.

In summary, the results presented in Table 2 indicate that the demand for foreign exchange derivatives at the end of 2002 had a strong speculative component.

3. Multivariate Analysis

In this section we study the determinants of the demand for currency swaps for the purpose of speculation or hedging. To do so, we follow Géczy, Minton and Schrand (1997) and Mian (1996) and estimate a logit model with the sample that we describe in the previous section composed of 343 firms. In the estimations of hedge, the dependent variable is equal to one for the firms that hedge and zero for those that do not. For the speculation estimations, the dependent variable is equal to one when the firm speculates and zero in the contrary. Next, we present the different control variables that we use in our regressions, grouped in accordance with diverse theoretical explanations both for hedging and for speculation.

3.1 Control Variables.

3.1.1 Costs of Bankruptcy

Smith and Stulz (1985) argue that a hedge is a method by means of which corporations can reduce the volatility of their cash flow. The choice to hedge occurs more frequently among firms with greater costs of bankruptcy or greater probabilities of bankruptcy.

However, a corporation with high leverage has a greater probability of bankruptcy. For an empirical approximation of the level of leverage we follow Géczy, Minton and Schrand (1997) and use the ratio of the accounting value of the long term debt to the size of the firm. This last variable is defined, as in Graham and Rogers (2002), as the logarithm of the volume of assets.

Export revenues and import expenses increase the currency exposure of corporations and thus can increase the probability of bankruptcy in the case of a currency appreciation or depreciation, respectively. As an empirical approximation for

11

export revenues and import expenses we use the ratio of the volume of external revenue to total gross revenue, and the ratio of the total importation expenses in relation to gross revenue respectively.

In the same manner, external debts, which result in mismatches between currencies of assets and currencies of liabilities, increase the firms' currency exposure and can imply that currency depreciations increase the probability of bankruptcy. For empirical approximations for a firm's external debt we use a binary variable equal to one if the business has dollar-denominated debt and zero otherwise; the ratio between the total short term external debt and the logarithm of total assets; and the ratio between the total external debt and the logarithm of total assets.

Finally, the ratio between current assets and current liabilities shows the degree of the firm's current liquidity. Extremely liquid businesses will have less incentive to hedge and greater incentive to speculate because, in this case, they have a lesser probability of bankruptcy.

3.1.2 Costs of Agency with Creditors

Myers (1977) demonstrates that indebted businesses have distorted incentives in terms of their policies for investment. To summarize, the distortion occurs due to the priority that the creditors have over the shareholders for receiving cash flow generated by corporations. Given this priority, the shareholders do not have incentives to contribute resources for investments whose returns – because of the highly indebted situation – will likely be used in the payment of debt. Excessive debt, however, can impede lucrative projects from being implemented. Thus, creditors anticipate the conflict of interest and incorporate their costs in the interest rate.

Mayers and Smith (1982) show that a hedge reduces the probability of a company not fulfilling its obligations, thus reducing the probability that the investments are distorted and, consequently, benefiting the shareholders through the reduction of the interest rate. Hedging, therefore, takes a firm's investment policy closer to that which maximizes the firm's value.

On the other hand Jensen and Meckling (1976) argue that business with great amounts of debt can choose excessively risky investments. Following this thread, Géczy, Minton and Schrand (1997) show that costs of agency with creditors can induce the businesses to speculate. This can occur if shareholders turn their shares into options above the value of a leveraged firm, speculating to increase the volatility of the firm's cash flow when close to bankruptcy.

We have, therefore, two conflicting forecasts. On the one hand, Mayers and Smith (1982) argue that corporations highly in debt are more likely to hedge. On the other hand, Géczy, Minton and Schrand (1997) argue that corporations with significant debt have greater incentive to speculate. In order to determine which of these effects prevail, we use two variables to capture costs of a suboptimal investment policy: the ratio between the total value of fixed assets and the size of the corporation and the ratio of the market value of corporation and its book value.

The higher the ratio between the fixed assets and the logarithm of total assets, the greater the firm's capacity to offer real collateral to creditors, that can reduce the creditors' loss due to financial stress and, consequently, reduce the incentives to distort the investment policy. Therefore, a greater ratio between fixed assets and the logarithm of the total assets reduces both the probability to hedge and to speculate.

In contrast, a high ratio between a corporation's market value and the book value suggests that future gains (embedded in the market value of the firm's shares) still do not correspond to the value of the existing assets. Such a corporation should have greater difficulty offering real collateral to creditors compatible with the profitability of the existing investment opportunities. Thus, we expect a positive relationship between the ratio of the market value and book value and the probability of hedging or speculation.

Another characteristic of a firm related to its cost of agency with creditors is its size. Larger firms, in general, have greater reputation, a fact that can reduce costs of agency. Therefore, we can expect that the size, defined as above, reduce the probability of the firm using hedge or speculation.

Nance, Smith and Smithson (1993) argue that corporations, by substituting debt for preferential shares, reduce the probability of bankruptcy, thus reducing the cost of agency, without the need to hedge. The authors anticipate a negative relationship between the volume of preferential shares and the probability of hedging. Géczy, Minton and Schrand (1997), on the contrary, argue that there is a positive relationship: firms with more financial restrictions tend to adopt a suboptimal investment policy. Because, according to the authors, the preferential shares increase the financial costs, the probability to hedge increases. In order to test these hypotheses, we include in our regressions the ratio between the book value of the preferential shares and the logarithm of the firm's total assets.

We also consider another explanatory variable that is related to both the costs of bankruptcy and to the cost of agency with creditors: the firm's profitability. The firm's profitability is defined as the ratio of the company's net revenue to its net worth. This variable gives an idea about the capacity of the corporation to internally finance itself, avoiding the capital market or bank loans. The less a company needs to finance externally, the less are the costs of bankruptcy and the less is the necessity to hedge. Or the company can run greater risks, for example, by speculating. On the contrary, more lucrative firms can be subject to greater costs related to investment policies because they have more available projects from which to choose, a fact that suggests a greater demand to hedge. This being the case, the impacts of profitability over the probabilities to hedge or speculate are uncertain.

3.1.3 Assymetric Information

De Marzo and Duffie (1991) suggest that corporations with greater asymmetry of information between executives and shareholders can obtain larger profits by hedging. De Marzo and Duffie are concerned with the shareholders' capacity to choose from their portfolios of assets. Hedging reduces the volatility of the companies' cash flow that, in turn, reduces the uncertainties of the shareholders' set of information. Consequently, the shareholders accept a hedge because this improves their portfolio choices. As an empirical approximation for asymmetric information between executives and shareholders we use the number of institutional investors of the firm. The idea is that institutional investors invest in the acquisition of information diminishing the uncertainty about the value of the firms in their portfolios. Therefore, a great number of institutional investors indicate a lesser probability of the firm performing a hedge.

3.1.4 Aversion of Executives or Shareholders to Risk

The volatility of their compensation imposes costs to executives or controllers contrary to risk. Stulz (1984) and Smith and Stulz (1985) argue that if the optimum

contracts for compensation of executives or controllers contrary to risk are related to the volatility of the corporation' revenue or cash flow, these volatilities can be costly for these agents. If the executives or controllers do not manage to hedge on their own, or if it is up to the corporations to choose to hedge, then a hedge done by the firm can increase the well being of the administrators. At the same time, Ljungqvist (1994) shows that, in firms severely in debt, the participation of executives or controllers in profits can serve as an incentive for them to speculate.

We use two variables as approximations for the executive's risk exposure: one binary variable equal to one if the executive has participation in profit and zero otherwise; and another variable that shows the executives total compensation. The participation in profits and executives compensation are obtained by the firms' financial statements provided by CVM¹⁰.

3.1.5 Taxes

Graham and Rogers (2002) discuss the impact of taxes on incentives for corporations to hedge. They examine two impacts. The first is related to the increase in the level of debt of the firm. In countries in which financial expenses imply a fiscal benefit, hedging increases value by increasing a firm's capacity for debt and, consequently, by allowing a lower tax payment.

A second fiscal incentive to hedge is related to the convexity of the expectation of tax payments. Mian (1996) presents evidence that the awaited payment of taxes is a convex function of the generation of a firm's cash. In this case, the Jensen's inequality shows that a hedge can reduce the expected payment of taxes.

In order to test for the impact of taxes on the decisions to hedge, we use a binary variable equal to one when the company pays taxes and zero otherwise. We expect that firms that pay taxes have are more likely to hedge.

3.1.6 Economies of Scale

Mian (1996) argues that risk management programs by means of derivatives can present initiation, implementation and maintenance costs. If these costs are significant, a

¹⁰ The empirical literature makes use of the total shares or of the total volume of options of shares of the corporations in the hands of the executives to study the relationship between the volatility of their compensation and the cash flow of corporations. Such variables are not available in Brazil.

company may not use these programs. Such costs present economies of scale related to the size of the firm. Therefore, the size of the firm – measured by the log of assets can be positively related to the probability of hedging *or* the probability of speculation.

3.1.7 Multinational Firms

In the regressions to study speculation as well as those to study hedge, we use as a control variable a binary variable equal to one if the firm is multinational and zero otherwise. Multinational corporations can be interested in protecting the interests of shareholders in the origin country against fluctuations of the nominal exchange rate, even without possessing export revenues, import expenses or debt in dollars. In this case, multinational firms should be more likely to use a hedge and less likely to speculate.

3.1.8 Privileged Information in the Foreign Currency Market

Corporations with revenues from exports or expenses from imports are natural candidates to speculate with foreign exchange derivatives. The nature of their activities makes these corporations follow regularly the foreign exchange market, maintaining close contact with agents that are probably the first to detect changes in the trends of the nominal exchange rate (*dealers* of foreign currency, for example). Therefore, they can participate in the foreign exchange market using privileged information. Our prior is that corporations, which have relative relevant export revenues or imports expenses, are more likely to speculate.

Additionally, we use in our regressions explanatory variables that indicate the participation of firms in sectors where export revenues or import costs are more relevant. These sectors are more affected by exchange rate fluctuations. To verify if this participation is important in order to explain a business' decision to hedge, we include binary variables equal to one when a business is part of one of these sectors, zero otherwise¹¹.

¹¹ We consider the concessionaries of public services in the group of firms with high currency exposure. One simple observation of the data we dispose is sufficient to make clear that a large number of corporations of this sector perform currency swap operations during our sample period and have debt in dollars.

3.2 Results

Table 3 presents Pearson's correlations of the diverse independent variables that we use in our estimations. Only three correlations are significant at the 5% level and are related to variable size, defined as the logarithm of a firm's total assets.

We now move on to the analysis of a firm's probability to hedge or to speculate. The results of the logit estimations of these probabilities are presented in Table 4. As the Table shows, the $\chi 2$ statistics of the test for the ratio of maximum likelihood does not reject the overall significance of the variables. Although they are not reported in the table, we also do several diagnosis tests for omitted variables. In particular, we include squared terms of the independent variables that are not dummy variables and verify that the coefficients of these additional variables are not collectively different from zero¹². These tests indicate a correct specification of the model.

Column A of Table 4 highlights the importance of the existence of external debt as a determining factor in the choice to hedge. The existence of external debt increases the probability of hedge by 3%. On the contrary, other factors related to operational currency exposure such as export revenues and import expenses are not significant. The first result is what we would expect.

The positive relationship between the existence of dollar-denominated debt and the probability to hedge is robust. Tests not described in the tables show corporations with debt in dollars of the most varied sectors, tradeables and non-tradeables, hedging, trying to prevent depreciations of the nominal exchange rate from negatively affecting their financial obligations and, consequently, their cash flows.

Economies of scale also affect positively the probability to hedge. Larger firms, for which the relative costs to start up and to maintain risk management programs by means of derivatives are relatively small, are more likely to hedge, as suggested by the positive coefficient for firm size. Therefore, since smaller companies have greater costs of bankruptcy and greater asymmetry of information, we have evidence in favor of the hypothesis of economies of scale and against hypotheses centered on asymmetric information and costs of bankruptcy.

¹² For example, for the mentioned estimation for the probability to hedge, the p-value of the test for omitted variables is 0.35, whereas for the estimation for the probability to speculate the p-value is 0.22.

The fact that a firm is multinational positively affects the probability to hedge. It was what we expect a *priori*, given that, as we have already argued, multinationals possess natural demand to hedge.

Although we do not show the estimated coefficients of the indicator variables of the industries, it is important to mention that the participation of the corporations in the majority of the economic sectors that we select is not relevant to explain the increase in the probability to hedge. Even the coefficient for the concessionaries of public service sector, which we suspected would have a greater tendency towards currency protection, does not appear to be significant. Of the export sectors, only the transportation sector negatively affects the disposition of the firms' to seek currency protection.

The data do not identify aversion of risk of executives or controllers as a determinant in the choice to hedge. The participation in profits is not significant in affecting the probability to hedge. The results are also not indicative that the level of firms' leverage affects significantly the decision to hedge, as postulated by the theory of cost of agency with creditors. In the same manner, the evidence does not support the models of asymmetric information or models related to taxes. Finally, none of the variable coefficients that suggest opportunities for growth are significant.

The results of the speculation analysis are presented in column B of Table 4. The results show that export revenues affect positively the probability to speculate. A 1% increase in the ratio of export revenues and gross revenues increases the probability of speculation by 5%. The existence of debt in dollars does not affect speculation. This is fairly reasonable; given that column A of Table 4 shows that external debt increases the probability to hedge.

Besides export revenues, only one other variable positively affects the probability of speculation: the participation of corporations in the transportation sector (results not shown in the table). This participation positively affects the speculative demand and is consistent with our univariate analyses, which show that the firms of this sector with open positions speculated, and with the results from the logit estimation of the demand to hedge, which show that the firms participation in this sector negatively affects the probability to hedge.

Concluding, we can confirm that, in 2002, the results of the logit estimations provide evidence that factors of currency exposure – debt in dollars and revenues from exports are the most important determinants of hedging and of currency speculation, respectively. Following, we show that these results are robust in different samples of open capital firms, in different control variables and other econometric techniques.

4. Analysing Hedge and Speculation in the Years from 1999 to 2001

In a first attempt to investigate the robustness of our results, we analyze whether the decisions to hedge or speculate are similar in other years, between 1999 and 2001. During this period, the volatility of the exchange rate was much lower than it was during 2002. Even in 1999, the currency crisis occurred in the beginning of the year and lasted for much less time than the currency crisis in 2002. Therefore, the incentives, especially for speculation, are quite different than those of 2002.

We look at open foreign exchange swaps at the end of each year. We consider the net position, long minus short in foreign exchange. We use again the same definition as before for hedge and speculation, reverse or neutral. The number of firms that hedge during this period is 21 in 1999, 32 in 2000 and 40 in 2001. The number of firms that speculate is 8 in 1999, 13 in 2000 and 15 in 2001. Of the latter, the number of firms that do reverse speculation is 7 in 1999, 4 in 2000 and 7 in 2001.

Initially, we make logit regressions for the years 1999 to 2001. The results to hedge, which we present in Panel A of Table 6 show once again that the presence of foreign exchange debt is the most important reason for companies to hedge in these periods. It is significant in all periods.

However, when we perform the logit regressions for speculation for these years, we observe as shown in Table 6 that the results change substantially from the ones we obtain in 2002. Export revenues divided by gross revenues are no more significant in explaining speculation.

The results indicate that the year 2002 was atypical in terms of incentives placed for speculation. Possessing privileged information in the market made exporters take advantage of these incentives in 2002. This does not occur in other years as the results of the logit estimations clearly demonstrate.

To verify how relevant these incentives are for speculation on the part of exporters, we turn to a balanced Panel analysis considering the whole sample period from 1999 to 2002 panel. We consider random effects¹³. We change slightly our basic model, including a dummy for the year 2002, the year of great volatility of the exchange rate, as well as interacting this dummy with the variable that measures export revenues.

The results are presented in Table 7. In the case of hedge, the results are very similar to those that we obtain in the logit estimation of the year 2002. Existence of debt and size are still positively related to the probability to hedge. The dummy 2002 is also significant, p-value of 0.05, which indicates the relevance of this year for hedge too.

In the case of speculation, we see that the variable for exports revenues divided by gross revenues affect the probability of speculation only in 2002, p-value of 0.09. The dummy for the year 2002 is highly significant, as well, p-value of 0.0. This confirms that the year 2002 is very much atypical as far as speculation on the part of firms that exports is concerned.

As Table 7 shows, the χ^2 statistic does not reject the overall significance of all of the independent variables in the model. Although we do not mention this in the tables, we once again perform several tests for omitted variables and redundant variables (squared terms of the non-dummy independent variables) that do not reject the specification of the model.

We also try several other specifications. In one, we include a dummy for 1999, where a foreign exchange crisis occurs at the beginning of the year, and another variable in which we interact it with the variable that measures exports revenues. Both variables are not significant. These results indicate once more that the incentives for speculation in 2002 are very strong and different from the incentives of speculation of other years.

¹³ We have a dummies as regressors which makes fixed effects not possible.

5. Testing the Robustness of Our results

5.1 Excluding Neutral Speculation

In our second attempt to test the robustness of the results, we will exclude the corporations that speculate in neutral form, or rather, those that do not have operational currency exposure, which we continue to define as the existence of export revenues, import expenses, or debt in dollars, but that still maintain open positions in currency swaps. As we argue earlier, these corporations are more likely to be incorrectly classified as speculators.

Table 5 presents the results of the logit estimations for the two possible alternatives in the sample that exclude neutral speculation: hedge and reverse speculation (currency exposures and open positions of swap of the same signal). As in our initial sample, the $\chi 2$ statistics of the tests of the ratio of maximum likelihood does not reject the combined significances of all the variables. We implement, once again, several tests for omitted variables (squared terms of the independent variables that are not dummy variables) that, though not presented in the tables, indicate a correct specification of the model.

Once again, debt in dollars and the size of the firm positively affect the probability of corporations to hedge. At the same time, external revenues positively affect the probability for speculation.

5.2 Endogeneity of Debt

One relevant critique of the empirical tests done so far is that some independent variables, which measure potential incentives to hedge or speculate, also can be choice variables. In particular, the variables that cause the greatest concern are those related to the cost of bankruptcy, this because the choice of capital structure, which affects the expected cost of financial stress, is a joint decision with the decision to hedge.

One way to minimize this problem is by simultaneously modeling the debt decisions and the decision to hedge, as do Géczy, Minton and Schrand (1997). Following these authors, we suppose that: the gross revenues and the ratio between the fixed assets and the logarithm of the firm's assets show the firm's ability to provide collateral, with the following increase of the capacity for debt; the existence of a hedge

increases the capacity for debt by diminishing the risk of financial stress; and finally, we use indicator variables for industries (one for each of the eight classifications of industry) as a way to control characteristics inherent to the actuation industry that could affect the cash flows of the creditors (for example, regulatory risks).

Finally, again following Géczy, Minton and Schrand (1997), we suppose that the decision to hedge is explained by the same independent variables of the logit estimations of Tables 4 and 5. We have, therefore, a system of two equations to be estimated: the equation for debt and the equation for the decision to hedge, which we estimate by minimum squares in two stages. Based on these estimations we test the restrictions of the coefficients of the models (Wald tests) that prove that, in fact, the two equations should be estimated simultaneously.

In order to save space, once again we do not present the complete results of the regressions. However, it is worth mentioning that, for the equation for debt, the expected estimated coefficient for hedge is not statistically significant and that, in the equation for the decision to hedge, the coefficient for debt is also not statistically significant. More importantly, however, is that, in the equation for the decision to hedge the same signal for the variables that also entered in the logit equation to hedge that ignored endogeneity of debt. Therefore, we can confirm that our empirical results about the decisions to hedge were not affected by problems of endogeneity of debt.

5.3 Future Flux of Exports Revenues and Imports Expenses

In the empirical analyses done so far, we consider only current export revenues and import expenses, that is, those related to the 2002 fiscal year. Nevertheless, it is possible that exporter or importer firms decide to speculate or hedge in a certain year taking into account the expected value of future fluxes of export revenues and import expenses. To analyze this possibility, we suppose that the expected future values of these revenues or expenses are equal to their 2002 values. We then define operational currency exposure as the present value of export revenues minus the sum of debt in dollar and the present value of import expenses. Note, however, that the positions of hedging, neutral speculation and reverse speculation continue to be defined by the sign of the product between the net position of open swaps and the operational currency exposure. We verify that the change in the definition of operational currency exposure does not alter the classification of hedging, neutral speculation and reverse speculation for any of the 93 corporations with open positions in currency swaps. However, the new definition of operational currency exposure modifies the classification of the positions of 9 firms of the control group of 250 businesses that we use. This difference forced us to re-estimate all of the logit estimations of Tables 4, 5, 6 and 7, substituting the present values of external revenues and imports for their future values in the regressions. The results not reported are qualitatively similar to those presented until now.

6. Conclusion

Géczy, Minton and Schrand (1997) show that, in 1990, 41.1% of North American firms pertaining to the Fortune 500 group use foreign exchange derivatives in the year 1990. Why do so many corporations demand foreign exchange derivatives?

In order to investigate this question, we build an original database made up of 25,457 contracts for open currency swaps open at the end of 2002. Looking at these contracts we identify 93 corporations with open position in foreign exchange swaps. Of these 93 corporations, 53 demanded swaps in order to hedge. Or rather, the contracts to swap reduce the firms' exposure to currency risk.

Of the 53 firms that purchase swaps in order to hedge, all hold long positions– gains with currency depreciation – 21 concentrated in the utilities sector and all have debt in dollars. In fact, the existence of external debt proves to be the principal determinant of hedge.

The data show, however, that various corporations demand swaps in 2002 for speculative purposes. Of 93 businesses with open positions in currency swaps at the end of 2002, 40 speculate. Of these, 18 speculate increasing the risk of their operational currency exposures. The other 20 speculate without having operational currency exposure. We also conclude that firms with larger export revenues are more likely to speculate.

In summary, this study suggests that in periods of great volatility of the exchange rate as in the year 2002 – the existence of debt in dollars is the principal

23

determinant of the demand to hedge, and that the firms' demand for foreign exchange derivatives is indeed related to speculative motives.

This paper contributes in a relevant way to the literature because by using a unique database made of swap contracts of corporations in Brazil it makes it possible to distinguish much better the incentives related to speculation and hedge in the foreign exchange market by corporations. As the literature shows the capacity to distinguish these incentives is rare due to the fact that previous research has had access only to off balance sheet information of corporations¹⁴. Future research should assess whether in the last few years firms have resorted to increased hedged of their cash-flows in dollars, either through derivatives or through "natural hedges", such as investing abroad.

¹⁴ See Tirole(2006) for a discussion of the problems related to off balance sheet information of corporations.

References

De Marzo, Peter and Duffie, Darrell. "Corporate Financial Hedging with Proprietary Information". Review of Financial Studies, 8, 1991, 743-771.

Géczy, Christopher, Minton, A. Bernadette and Schrand, Catherine. "Why Firms Use Currency Derivatives". Journal of Finance, 52, 1997,1323-1353.

Graham, R. John and Rogers, A. Daniel. "Do Firms Hedge in Response to Tax Incentives?" Journal of Finance, 57, 2002, 815-838.

Jensen, C. Michael and Meckling, H. William. "Theory of the Firm: Managerial Behavior, Agency Costs and Ownership Structure". Journal of Financial Economics, 3, 1976, 305-360

Ljungqvist, Lars. "Asymmetric Information: A Rationale for Corporate Speculation". Journal of Financial Intermediation, 3, 1994, 188-203.

Mayers, David. "Why Firms Issue Convertible Bonds: The Matching of Financial and Real Options". Journal of Financial Economics, 47, 1998, 83-102.

Mayers, David and Smith, W. Clifford. "On the Corporate Demand for Insurance". Journal of Business, 55, 1982, 281-296.

Merton, Robert. "A Simple Model of Capital Market Equilibrium with Incomplete Information". Journal of Finance, 42, 1987, 483-510.

Mian, L. Shehzad. "Evidence on Corporate Hedging Policy". Journal of Financial and Quantitative Analysis, 31, 1996, 419-439.

Myers, S. "Determinants of Corporate Borrowing", Journal of Financial Economics, 3, 1977, 147 – 175.

Nance, D. R., C. W. Smith and C. W. Smithson. "On the Determinants of Corporate Hedging". Journal of Finance, 48, 1993, 267-284.

Saito, Richard and Schiozer, F. Rafael. "Why do Latin American Manage Currency Risks?" 2006, in Mimeo.

Smith, W. Clifford and Stulz, M. René. "The Determinants of Firms Hedging Policies". Journal of Financial and Quantitative Analysis, 20, 1985, 391-405.

Stulz, M. René. "Optimal Hedging Policies". Journal of Financial and Quantitative Analysis", 19, 1984, 127-140.

Tirole, Jean. "The Theory of Corporate Finance", Princeton University Press, 2006.

Table 1. Financial Characteristics

This table presents the financial and accounting characteristics of the firms that demand currency swaps and of those that do not. The sample of the firms is formed by a group of 343 non-financial open capital corporations. They are not of the same economic group, divulged all of the necessary accounting information and have some form of exchange rate exposure. All of the information is from the end of the 2002 fiscal year, and relative to the financial statements that are available at CVM. The existence of external debt is equal to one when a firm possesses debt in dollars and zero otherwise; the participation of administrators in profit is equal to one when administrators participate in profit and zero otherwise; tax is equal to one when the firm pays tax and zero otherwise; institutional investors show the number of institutional investors of the firm; preferential shares are the book value of the firm's preferential shares. The number of observations of each characteristic is in the second column (N). The t statistics are presented for the average test between financial characteristics of the firms' diverse categories. The tests suppose equal variances unless the hypothesis is rejected at 5%. The p-values are in parentheses.

Panel A: Firms with and without Open Positions in Currency Swaps

Financial Characteristics		Firms with open positions in currency swaps at the end of 2002 (A)			Firms without open positions in currency swaps at the end of 2002 (B)				Average Tests
	Ν	Average	Median	Standard Deviation	Ν	Average	Median	Standard Deviation	A-B
Log(Assets)	93	13.43	13.98	2.74	250	12.14	12.70	3.42	3.19 (0.0)
Existence of External Debt	93	0.68	1.0	0.46	250	0.45	0.0	0.49	0.23 (0.0)
Export Revenues/ Gross Revenues	93	7.7	1.24	49.1	250	4.4	1.88	20.5	3.3 (0.05)
Import Expenses/Gross Revenues	93	1.79	1.04	13.10	250	1.25	0.44	10.8	0.54 (0.18)
Administrators Participation in Profits	93	0.53	0.98	0.49	250	0,38	0,0	0,48	0,25 (0,0)
Administrators Compensation (R\$mil)	93	239,56	250	14897	250	28031	0,0	276500	-4417 (0,0)
Tax	93	0,49	0	0,50	250	0,74	1,0	0,43	-0,25 (0,0)
Fixed Assets/ Log(Assets)	93	142814	3500	630.777	250	192500	27000	800.351	49686 (0.55)
Market Value/Book Value	93	0,56	0,0	1.33	250	0,29	0,0	2,64	0,27 (0,35)
Number of Institutional Investors	93	80,29	0,0	280,78	250	1142	0,0	17129	-1062,20 (0,55)
Preferential Shares /log(Assets)	93	52192	0,0	257100	250	10919	0,0	63186	41273 (0,43)
Long Term Debt/Log(Assets)	93	274019	23454	158521	250	183374	5600	10800	92030 (0.38)
Total Dollar Debt/Log(Assets)	93	29337	0,0	64595	250	14834	0,0	99050	14335 (0,32)
Short Term Dollar Debt/Log(Assets)	93	34640	0,0	199383	250	16552	4050	34325	18058 (0,49)

Table 2. Hedge or Speculation

Panel A presents the corporations with open contracts of currency swaps in 2002 by type of position: hedge, reverse speculation and neutral speculation. The firm hedges when the product between its operational currency exposure, defined as the difference between export revenues and the sum of import expenses and debt in dollars, and its open positions in currency swaps is less than zero. The firm speculates when the product between its operational currency exposure, defined as the difference between export revenues and the sum of import expenses and debt in dollars, and its open positions in currency exposure, defined as the difference between export revenues and the sum of import expenses and debt in dollars, and its open positions in currency swaps is greater than zero. The speculation is reverse when the product of the open positions in currency swaps and the operational currency exposure, but possesses open positions in currency swaps. Swaps/ NW is the average level in the industry of the ratio of net positions in currency swaps to net worth. Panel B presents the firms that hedge and those that perform reverse speculation classified by sectors; the form in which they do so and if they possess debt in dollars or not. Panel C presents the firms that hedge and those that perform in which they do so and if they have exports or imports in dollars. Panel D shows the origin of the corporations that hedge, or that perform reverse or neutral speculation.

		Hedge		Revers	e Specu	llation	Neutra	I Specul	ation	
Industries	Net Position		Swaps/ NW	Net Position		Swaps/ NW	Net Position		Swaps/ NW	Total
Chemical and Petroleum	9	0	0,05	2	0	0,03	1	0	0,04	12
Food and Beverages	1	0	0,05	7	0	0,04	2	0	0,05	10
Mining and Metallurgy	2	0	0,01	0	0	N.A.	5	0	0,06	7
Electro/Electronic Equipment	6	0	0,08	5	0	0.026	4	0	0,07	15
Transportation	3	0	0,05	0	0	N.A	0	0	N.A.	3
Concessionaries for Public Service	19	0	0,06	0	0	N.A	6	0	0,015	25
Textiles	5	0	0,04	0	0	N.A	5	0	0,03	10
Metallurgy	6	0	0,07	2	0	0,03	0	0	0,12	8
Other	2	0	0,04	0	0	N.A	1	0	0,047	3
Total (long + short)	53			16			24			93

Panel A: Types of Positions and Currency Exposures

		Hec	lge		Reverse Speculation				
Industries	With	Debt	Withou	t Debt	With	Debt	Withou	t Debt	
	Short	Long	Short	Long	Short	Long	Short	Long	Total
Chemical and Petroleum	9	0	0	0	1	0	1	0	11
Food and Beverages	1	0	0	0	4	0	3	0	8
Mining and Metallurgy	2	0	0	0	0	0	2	0	4
Electro/Electronic Equiptment	6	0	0	0	3	0	2	0	11
Transportation	3	0	0	0	0	0	0	0	3
Concessionaries of Public Service	19	0	0	0	0	0	0	0	19
Textiles	5	0	0	0	0	0	0	0	5
Metallurgy	6	0	0	0	2	0	0	0	8
Other	2	0	0	0	0	0	0	0	2
Total (with debt + without debt)	53	3	0	1	10)	0		63

Panel B: Hedge and Reverse Speculation – With and Without Debt in Dollar

	Hedge				Reverse Speculation				
Industries	Exp	orts	Imp	orts	Expo	orts	Imp	orts	Total
	Long	Short	Long	Short	Long	Short	Long	Short	
Chemical and Petroleum	4	0	3	0	1	0	0	0	8
Food and Beverages	0	0	0	0	7	0	4	0	11
Mining and Metallurgy	0	0	0	0	0	0	0	0	0
Electro/Electronic Equipment	3	0	3	0	5	0	2	0	13
Transportation	1	0	0	0	0	0	0	0	1
Concessionaries for Public Service	0	0	0	0	0	0	0	0	0
Textiles	3	0	2	0	0	0	0	0	0
Metallurgy	6	0	6	0	2	0	2	0	16
Other	0	0	0	0	0	0	0	0	0
Total (exports + imports)	14	4	1,	4	1	5	8	}	41

Panel C: Hedge and Reverse Speculation - Exports and Imports

Panel D: Hedge and Speculation – Capital Origin

Origin of Capital	Hedge	Reverse Speculation	Neutral Speculation
Private Domestic	44	13	22
State-owned	5	1	1
Multinational	4	2	1
Total	53	16	24

Pearson correlations equal to one when a in profit and zero otl preferential shares au	for the rel firm poss herwise; ta re the book	gressor sesses d ax is eq k value	s used in the lc ebt in dollars ; lual to one whe of the preferer	git estimations and zero otherv en a firm pays ntial shares of t	. The correls vise; the par tax and zero he firm.	ations ticipati otherv	in black al lon of adm vise; instit	nd in paren ninistrators utional in	theses are in profits vestors sho	significant at 5%. is equal to one whe number of ii	The existention of the adminication of the adm	ce of exte nistrators nvestors	rnal debt is participate of the firm;
	Exis Exte Deb	stence of ernal ot	Exportation/Gross Revenue	Importation/Gross Revenue	Circulating Asset/ Circulating Liability	Тах	Investors	Long Term Debt/Log (assets)	Immobilized/ Log(assets)	Log(assets) Market Value/Book Value	Participation in Profits	Profitability	Total Debt in Dollars/Assets
Exportation/Gross Revenue	0,15												
Importation/Gross Revenue	-0,08 0,0	Q											
Circulating Assest/Circulating Liability	-0,04 0,0	ß	-0,01										
Tax	-0,05 0,0	2	0,01	0,01									
Institutional Investors	0,06 0,0	Q	0,07	-0,01	0,07								
Long Term Debt/Log (assets)	-0,04 0,0	Ξ	-0,01	-0,02	-0,11	-0,01							
Immobilized/Log(assets)	0,08 0,0	71	-0,01	-0,02	0,00	-0,01	0,28						
Log (ativos)	(0,34) 0,0	2	-0,07	-0,09	-0,01	0,04	(0,4)	(0,33)					
Market Value/Book Value	-0,01 0,0	8	0,08	0,07	0,12	0,01	-0,02	0,00	0,05				
Participation of Administrators in Profits	0,12 0,0	-	-0,02	-0,01	0,07	0,08	-0,05	-0,05	0,03	0,01			
Profitability	0,07 0,0	4	-0,02	-0,01	0,06	0,00	0,01	0,03	0,08	0,02 0,07			
Total Debt in Dollars / log(assets)	-0,03 0,1	-	-0,01	-0,01	0,06	0,40	-0,02	-0,01	0,06	-0,01 0,11	-0,01	-0,01	-0,02

Table 4. Logit Estimations: Firms with Currency Exposure

Logit regressions of firms probability to hedge and to speculate. The sample includes 343 corporations that we select from the financial statements of 2002 and that are subject to some type of risk related to currency exposure. These corporations either have export revenue in dollars, import expenses in dollars, debt in dollars, presented open positions in currency swaps or take part in industrial sectors that are most affected by the currency exposure. Operational currency exposure is the difference between revenue from exports and the sum of expenses from imports and debt in dollars. The corporations that hedge are those for which the product of its operational currency exposure by the net open position in currency swaps at the end of 2002 is less than zero. The firms speculate if the product of its operational currency exposures with the net open positions in swaps at the end of 2002 is greater than zero or if they possess net open positions in swaps, but do not possess operational currency exposure. The existence of external debt is equal to one when a firm possesses debt in dollar and zero otherwise; the participation of the administrators in profits is equal to one when the administrators participate in profits and zero otherwise; tax is equal to one when the corporation pays tax and zero otherwise; institutional investors shows the number of institutional investors of the firm; preferential shares are the book value of the firm's preferential shares; and multinational is equal to one when a firm is multinational and zero otherwise. We are also controlling for the indicative variables of the following industrial sectors: Food/Beverage, Chemical/Petroleum, Metallurgy, Transportation, Mining, Electro/Electronic, Textiles and Concessionaries for Public Service. The statistics of the tests for maximum likelihood tests the joint significance of the variables. The robust standard errors are calculated using Huber-White. Below the estimated coefficients and the χ^2 , statistics, in parentheses, are the p-values.

Regressors	Dependent Variable Hedge (A)	Speculation (B)
Constant	-10.03 (0.0)	-1.01 (0.36)
Existence of External Debt	1.14 (0.03)	0.30 (0.47)
Log (Assets)	0.62 (0.01)	-0.087 (0.48)
Exports Revenues/Gross Revenues	-0.06 (0.10)	0.012 (0.0)
Imports Revenues/Gross Revenues	-0.05 (0.09)	0.03 (0.37)
Administrator Participation in Profits	-0.001 (0.10)	0.0 (0.17)
Compensation of Executives	-0.006 (0.06)	-0.0005 (0.80)
Tax	1.62 (0.19)	0.22 (0.73)
Long Term Debt/Log(Assets)	-0.002 (0.21)	0.0 (0.14)
Current Assets/Current Liabilities	0.008 (0.22)	-0.015 (0.45)
Fixed Assets/Log(Assets)	0.002 (0.86)	-0.0061 (0.57)
Market Value/Book Value	0.0 (0.11)	0.0 (0.0)
Preferential Shares /Log(Assets)	0.0002 (0.19)	0.024 (0.77)
Net Profit/ Total Equity	-0.05 (0.95)	-0.86 (0.52)
Multinational	0.0097 (0.03)	0.032 (0.23)
Institutional Investors	-0.001 (0.90)	-0.01 (0.57)
Dummies for Industries	controlled	controlled
Maximum Likelihood Ratio χ^2 (22)	58.80 (0.0)	53.57 (0.0)
Pseudo R ²	0.22	0.13

Table 5: Logit Estimations for Hedge and Reverse Speculation

Logit regressions of the firm's probability to hedge and probability of reverse speculation. Reverse speculation is defined as the positive product of the open position in currency swaps and the operational currency exposure, defined as the difference between export revenues and the sum of debt in dollars the import expenses. The corporations that hedge are those for which the product of their operational currency exposure by the net position in currency swaps was less than zero. The sample excludes the corporations that do neutral speculation. There are 323 firms with currency exposure, or rather, firms that have revenue in dollars, imports expenses in dollars, debt in dollars or participate in a tradable sector. The existence of external debt is equal to one when a firm possesses debt in dollar and zero otherwise; the participation of administrators in profits is equal to one when the administrators participate in profits and zero otherwise; tax is equal to one when the firm pays tax and zero otherwise; institutional investors shows the number of institutional investors of the firm; preferential shares are the book value of the firm's preferential shares; multinational is equal to one when a firm is multinational and zero otherwise. We are also controlling for the following industrial sectors: Food/Beverage, Chemical/Petroleum, Metallurgy, Transportation, Mining, Textiles, and Concessionaries for Public Service. The statistics of the tests for maximum verisimilitude and for the Lagrange multiplier test the combined significance of the dependent variables. The robust standard errors were calculated using Huber-White. Below the estimated coefficients and the χ^2 statistics, in parentheses, are the p-values.

_	Dependent Variable	
Hegressors	Hedge(A)	Speculation(B)
Constant	-2.45 (0.0)	-3.61 (0.0)
Existence of External Debt	0.98 (0.08)	1.24 (0.10)
Log (Assets)	0.13 (0.05)	0.94 (0.06)
Exports Revenues/Gross Revenues	0.29 (0.89)	0.85 (0.05)
Imports Revenues/Gross Revenues	-0.94 (0.54)	0.093 (0.54)
Administrator Participation in Profits	-0.99 (0.26)	0.027 (0.0)
Compensation of Executives	0.85 (0.07)	-0.01 (0.82)
Tax	-0.63 (0.37)	-0.32 (0.06)
Long Term Debt/Log(Assets)	0.41 (0.20)	-0.15 (0.34)
Current Assets/Current Liabilities	-0.01 (0.24)	-0.014 (0.24)
Fixed Assets/Log(Assets)	0.01 (0.0)	-0.15 (0.09)
Market Value/Book Value	0.57 (0.50)	0.056 (0.61)
Preferential Shares /Log(Assets)	-0.84 (0.69)	0.18 (0.07)
Net Profit/ Total Equity	0.02 (0.74)	-0.04 (0.0)
Multinational	0.38 (0.62)	0.47 (0.84)
Institutional Investors	0.45 (0.22)	0.90 (0.57)
Dummies for Industries	controlled	controlled
Maximum Likelihood Ratio χ^2 (22)	28.74 (0.18)	43.01 (0.0)
Pseudo R ²	0.12	0.28

Table 6: Logit Estimations of Hedge and Speculation for Other Years: 1999, 2000 and 2001

Logit regression for hedge and speculation for the years from 1999 to 2001. We use the complete sample of 343 firms. This sample includes all of the corporations that we select from the financial statements at the end of these years and that present some form of currency exposure. Panel A shows the logit estimations for firms that hedge. A firm hedges if it possesses open positions in currency swaps and operational currency exposure, defined as the difference between export revenues and the sum of import expenses and debt in dollars, and if the product of this exposure and of the positions is less than zero. Pannel B shows the logit estimations for firms that speculate. A firm speculates if it possesses open positions in currency swaps and operational currency swaps and operational currency exposure, defined as the difference between export revenues and the sum of import expenses and debt in dollars, and if the product of this exposure and of the positions for firms that speculate. A firm speculates if it possesses open positions in currency swaps and operational currency exposure, defined as the difference between export revenues and the sum of import expenses and debt in dollars, and if the product of this exposure and of the positions is greater than zero. The existence of external debt is equal to one when a firm possesses debt in dollars in profits is equal to zero when administrators participate in profits and zero otherwise; tax is equal to one when a firm pays tax and zero otherwise; multinational investors shows the number of institutional investors of the firm; preferential shares; multinational is equal to one when a firm is multinational and zero otherwise. We are also controlling for the following industrial sectors: Food/Beverage, Chemical/Petroleum, Metallurgy, Transportation, Mining, Textiles and Concessionaries for Public Service. The robust standard errors were calculated using Huber-White. We present below the coefficients and the χ^2 statistics, in parentheses, the p-values.

Panel A: Hedge for the Years 1999, 2000 and 2001

Regressors	1999	2000	20001
	(A)	(B)	(C)
Constant	-8.99	-4.69	-10.00
	(0.01)	(0.0)	(0.0)
Existence of External Debt	0.58	1.57	1.43
	(0.05)	(0.07)	(0.06)
Log (Assets)	0.65	0.14	0.54
	(0.03)	(0.20)	(0.01)
Exports Revenues/Gross Revenues	-0.002	-0.20	-0.23
	(0.24)	(0.80)	(0.27)
Imports Revenues/Gross Revenues	0.27 (0.88)	0.93 (0.94)	0.44 (0.21)
Administrator Participation in Profits	0.034	0.58	-1.29
	(0.13)	(0.21)	(0.29)
Compensation of Executives	-0.058	0.0	-0.011
	(0.06)	(0.21)	(0.08)
Tax	-0.25	-0.47	-0.26
	(0.29)	(0.09)	(0.56)
Long Term Debt/Log(Assets)	-4.31	0.03	0.83
	(0.11)	(0.52)	(0.91)
Current Assets/Current Liabilities	0.66	-0.001	0.49
	(0.71)	(0.91)	(0.74)
Fixed Assets/Log(Assets)	-0.81	0.13	1.20
	(0.58)	(0.87)	(0.41)
Market Value/Book Value	-111	-0.31	-0.04
	(0.22)	(0.46)	(0.91)
Preferential Shares /log(Assets)	0.92	0.42	-0.94
	(0.83)	(0.32)	(0.47)
Net Profit/ Total Equity	-0.001	-0.0013	-0.007
	(0.99)	(0.91)	(0.59)
Multinational	0.72	0.073	2.70
	(0.58)	(0.92)	(0.02)
Institutional Investors	0.01	-0.068	0.001
	(0.09)	(0.87)	(0.22)
Dummies for Industries	controlled	controlled	controlled
Maximum Likelihood Ratio χ^2 (22)	(0.24)	(0.01)	(0.0)
Pseudo R ²	0.24	0.21	0.26

Panel B: Speculation for the Years 1999, 2000 and 2001

Regressors	1999	2000	20001
	(A)	(B)	(C)
	-4,03 (0.05)	-6,3 (0.05)	-1,32
Constant	(0,05)	(0,03)	(0,20)
Existence of External Debt	(0,0)	-0,43 (0,53)	-2,42 (0,0)
Log (Assets)	-1,80	9,50	-1,47
	(0,77)	(0,0)	(0,85)
Exports Revenues/Gross Revenues	0,44	-11,79	-4,05
	(0,93)	(0,49)	(0,65)
Imports Revenues/Gross Revenues	0,51	0,97	0,63
	(0,21)	(0,11)	(0,15)
Administrator Participation in Profits	0,25	0,54	0,12
	(0,28)	(0,13)	(0,53)
Compensation of Executives	0,0	0,0	0,0
	(0,83)	(0,78)	(0,52)
Tax	-0,5	-0,17	-1,74
	(0,19)	(0,79)	(0,0)
Long Term Debt/Log(Assets)	0,0	0,0	0,0
	(0,88)	(0,66)	(0,71)
Current Assets/Current Liability	0,0	0,0	0,0
	(0,86)	(0,84)	(0,74)
Fixed Assets/Log(Assets)	0,0	0,0	0,0
	(0,57)	(0,66)	(0,88)
Market Value/Book Value	0,033	0,07	0,04
	(0,99)	(0,42)	(0,62)
Preferential Shares /log(Assets)	0,0	0,0	0,0
	(0,93)	(0,96)	(0,11)
Net Profit/ Total Equity	0,002	0,004	0,0
	(0,99)	(0,99)	(0,99)
Multinational	-0,99	1,99	1,38
	(0,47)	(0,08)	(0,30)
Institutional Investors	0,0	0,0	0,0
	(0,77)	(0,99)	(0,98)
Dummies for Industries	controlled	controlled	controlled
Maximum Likelihood Ratio $\chi^2(22)$	50.80	43.13	27.55
	(0.0)	(0.0)	(0.02)
Pseudo R ²	0.19	0.23	0.28

Table 7: Random Effect Panel Analysis for the Period 1999 to 2002: Firms with Currency Exposure

Pannel regressions of the probability of firms to hedge and the probability of the firm to speculate. The sample includes 343 corporations that we select from the financial statements from 1999 to 2002 and that are subject to some type of risk related to currency exposure. These firms have revenue in dollars, imports expenses in dollars, debt in dollars, present open positions in currency swaps or pertain to industrial sectors that are most affected by currency exposure. Operational currency exposure is the difference between exports revenues and the sum of import expenses and debt in dollars. We are only considering firms with net open positions in currency swaps purchased in dollars. The firms that hedge are the ones which the product of their operational currency exposure with their net open position in currency swaps at the end of 2002 is less than zero. The firms speculate if the product of their operational currency exposures with their net open positions in swaps at the end of each one of these years greater than zero or if they possess net open positions but do not possess operational currency exposure. The existence of external debt is equal to one when a firm possesses debt in dollars and zero otherwise; the participation of administrators in profits is equal to zero when administrators participate in profits and zero otherwise; tax is equal to one when a firm pays tax and zero otherwise; institutional investors shows the number of institutional investors of the firm; preferential shares are the book value of the firm's preferential shares; multinational is equal to one when a firm is multinational and zero otherwise. We are also controlling for the indicative variables of the following industrial sectors: Food/Beverage, Chemical/Petroleum, Metallurgy, Transportation, Mining, Electro/Electronic, Textile and Concessionaries of Public Service. This is a variable effect panel. The maximum likelihood statistic is presented. The robust standard errors are calculated using Huber-White. Below the estimated coefficients and the χ^2 statistics, in parentheses, are the p-values.

Regressors	Dependent Variable Hedge (A)	Speculation (B)
Constant	-16.64 (0.54)	-13.40 (0.18)
Existence of External Debt	0.87 (0.09)	0.35 (0.57)
Log (Assets)	1.48 (0.0)	2.36 (0.02)
Exports Revenues/Gross Revenues	-0.02 (0.34)	-0.007 (0.91)
Imports Revenues/Gross Revenues	-0.045 (0.69)	-0.0001 (0.65)
Administrator Participation in Profits	-0.18 (0.56)	0.17 (0.40)
Compensation of Administrators	0.0 (0.46)	0.0 (0.91)
Tax	0.07 (0.26)	0.0 (0.79)
Long Term Debt/Log(Assets)	-0.38 (0.87)	-0.4 (0.21)
Current Assets/Current Liabilities	0.58 (0.50)	0.31 (0.23)
Fixed Assets/Log(Assets)	1.48 (0.76)	0.31 (0.82)
Market Value/Book Value	0.01 (0.60)	-0.003 (0.11)
Preferential Shares /log(Assets)	0.13 (0.51)	0.05 (0.09)
Net Profit/ Total Equity	-0.098 (0.42)	-0.024 (0.01)
Institutional Investors	0.47 (0.82)	0.0046 (0.09)
D2002(ExportsRevenues/Gross Revenues)	-0.002 (0.34)	0.69 (0.09)
D2002 Dummies for Industries	0.75 (0.05) controlled	1.84 (0.0) controlled
Maximum Likelihood Ratio $\chi^2(15)$	-223.72 (0.03)	-118.03 (0.0)

Banco Central do Brasil

Trabalhos para Discussão

Os Trabalhos para Discussão podem ser acessados na internet, no formato PDF, no endereço: http://www.bc.gov.br

Working Paper Series

Working Papers in PDF format can be downloaded from: http://www.bc.gov.br

1	Implementing Inflation Targeting in Brazil Joel Bogdanski, Alexandre Antonio Tombini and Sérgio Ribeiro da Costa Werlang	Jul/2000
2	Política Monetária e Supervisão do Sistema Financeiro Nacional no Banco Central do Brasil Eduardo Lundberg	Jul/2000
	Monetary Policy and Banking Supervision Functions on the Central Bank <i>Eduardo Lundberg</i>	Jul/2000
3	Private Sector Participation: a Theoretical Justification of the Brazilian Position <i>Sérgio Ribeiro da Costa Werlang</i>	Jul/2000
4	An Information Theory Approach to the Aggregation of Log-Linear Models <i>Pedro H. Albuquerque</i>	Jul/2000
5	The Pass-Through from Depreciation to Inflation: a Panel Study Ilan Goldfajn and Sérgio Ribeiro da Costa Werlang	Jul/2000
6	Optimal Interest Rate Rules in Inflation Targeting Frameworks José Alvaro Rodrigues Neto, Fabio Araújo and Marta Baltar J. Moreira	Jul/2000
7	Leading Indicators of Inflation for Brazil Marcelle Chauvet	Sep/2000
8	The Correlation Matrix of the Brazilian Central Bank's Standard Model for Interest Rate Market Risk José Alvaro Rodrigues Neto	Sep/2000
9	Estimating Exchange Market Pressure and Intervention Activity <i>Emanuel-Werner Kohlscheen</i>	Nov/2000
10	Análise do Financiamento Externo a uma Pequena Economia Aplicação da Teoria do Prêmio Monetário ao Caso Brasileiro: 1991–1998 Carlos Hamilton Vasconcelos Araújo e Renato Galvão Flôres Júnior	Mar/2001
11	A Note on the Efficient Estimation of Inflation in Brazil <i>Michael F. Bryan and Stephen G. Cecchetti</i>	Mar/2001
12	A Test of Competition in Brazilian Banking Márcio I. Nakane	Mar/2001

13	Modelos de Previsão de Insolvência Bancária no Brasil Marcio Magalhães Janot	Mar/2001
14	Evaluating Core Inflation Measures for Brazil Francisco Marcos Rodrigues Figueiredo	Mar/2001
15	Is It Worth Tracking Dollar/Real Implied Volatility? Sandro Canesso de Andrade and Benjamin Miranda Tabak	Mar/2001
16	Avaliação das Projeções do Modelo Estrutural do Banco Central do Brasil para a Taxa de Variação do IPCA Sergio Afonso Lago Alves	Mar/2001
	Evaluation of the Central Bank of Brazil Structural Model's Inflation Forecasts in an Inflation Targeting Framework Sergio Afonso Lago Alves	Jul/2001
17	Estimando o Produto Potencial Brasileiro: uma Abordagem de Função de Produção <i>Tito Nícias Teixeira da Silva Filho</i>	Abr/2001
	Estimating Brazilian Potential Output: a Production Function Approach <i>Tito Nícias Teixeira da Silva Filho</i>	Aug/2002
18	A Simple Model for Inflation Targeting in Brazil Paulo Springer de Freitas and Marcelo Kfoury Muinhos	Apr/2001
19	Uncovered Interest Parity with Fundamentals: a Brazilian Exchange Rate Forecast Model <i>Marcelo Kfoury Muinhos, Paulo Springer de Freitas and Fabio Araújo</i>	May/2001
20	Credit Channel without the LM Curve Victorio Y. T. Chu and Márcio I. Nakane	May/2001
21	Os Impactos Econômicos da CPMF: Teoria e Evidência <i>Pedro H. Albuquerque</i>	Jun/2001
22	Decentralized Portfolio Management Paulo Coutinho and Benjamin Miranda Tabak	Jun/2001
23	Os Efeitos da CPMF sobre a Intermediação Financeira Sérgio Mikio Koyama e Márcio I. Nakane	Jul/2001
24	Inflation Targeting in Brazil: Shocks, Backward-Looking Prices, and IMF Conditionality Joel Bogdanski, Paulo Springer de Freitas, Ilan Goldfajn and Alexandre Antonio Tombini	Aug/2001
25	Inflation Targeting in Brazil: Reviewing Two Years of Monetary Policy 1999/00 Pedro Fachada	Aug/2001
26	Inflation Targeting in an Open Financially Integrated Emerging Economy: the Case of Brazil Marcelo Kfoury Muinhos	Aug/2001
27	Complementaridade e Fungibilidade dos Fluxos de Capitais Internacionais Carlos Hamilton Vasconcelos Araújo e Renato Galvão Flôres Júnior	Set/2001

28	Regras Monetárias e Dinâmica Macroeconômica no Brasil: uma Abordagem de Expectativas Racionais <i>Marco Antonio Bonomo e Ricardo D. Brito</i>	Nov/2001
29	Using a Money Demand Model to Evaluate Monetary Policies in Brazil Pedro H. Albuquerque and Solange Gouvêa	Nov/2001
30	Testing the Expectations Hypothesis in the Brazilian Term Structure of Interest Rates <i>Benjamin Miranda Tabak and Sandro Canesso de Andrade</i>	Nov/2001
31	Algumas Considerações sobre a Sazonalidade no IPCA Francisco Marcos R. Figueiredo e Roberta Blass Staub	Nov/2001
32	Crises Cambiais e Ataques Especulativos no Brasil <i>Mauro Costa Miranda</i>	Nov/2001
33	Monetary Policy and Inflation in Brazil (1975-2000): a VAR Estimation André Minella	Nov/2001
34	Constrained Discretion and Collective Action Problems: Reflections on the Resolution of International Financial Crises <i>Arminio Fraga and Daniel Luiz Gleizer</i>	Nov/2001
35	Uma Definição Operacional de Estabilidade de Preços <i>Tito Nícias Teixeira da Silva Filho</i>	Dez/2001
36	Can Emerging Markets Float? Should They Inflation Target? <i>Barry Eichengreen</i>	Feb/2002
37	Monetary Policy in Brazil: Remarks on the Inflation Targeting Regime, Public Debt Management and Open Market Operations Luiz Fernando Figueiredo, Pedro Fachada and Sérgio Goldenstein	Mar/2002
38	Volatilidade Implícita e Antecipação de Eventos de <i>Stress</i> : um Teste para o Mercado Brasileiro <i>Frederico Pechir Gomes</i>	Mar/2002
39	Opções sobre Dólar Comercial e Expectativas a Respeito do Comportamento da Taxa de Câmbio <i>Paulo Castor de Castro</i>	Mar/2002
40	Speculative Attacks on Debts, Dollarization and Optimum Currency Areas <i>Aloisio Araujo and Márcia Leon</i>	Apr/2002
41	Mudanças de Regime no Câmbio Brasileiro Carlos Hamilton V. Araújo e Getúlio B. da Silveira Filho	Jun/2002
42	Modelo Estrutural com Setor Externo: Endogenização do Prêmio de Risco e do Câmbio Marcelo Kfoury Muinhos, Sérgio Afonso Lago Alves e Gil Riella	Jun/2002
43	The Effects of the Brazilian ADRs Program on Domestic Market Efficiency <i>Benjamin Miranda Tabak and Eduardo José Araújo Lima</i>	Jun/2002

44	Estrutura Competitiva, Produtividade Industrial e Liberação Comercial no Brasil <i>Pedro Cavalcanti Ferreira e Osmani Teixeira de Carvalho Guillén</i>	Jun/2002
45	Optimal Monetary Policy, Gains from Commitment, and Inflation Persistence <i>André Minella</i>	Aug/2002
46	The Determinants of Bank Interest Spread in Brazil Tarsila Segalla Afanasieff, Priscilla Maria Villa Lhacer and Márcio I. Nakane	Aug/2002
47	Indicadores Derivados de Agregados Monetários Fernando de Aquino Fonseca Neto e José Albuquerque Júnior	Set/2002
48	Should Government Smooth Exchange Rate Risk? Ilan Goldfajn and Marcos Antonio Silveira	Sep/2002
49	Desenvolvimento do Sistema Financeiro e Crescimento Econômico no Brasil: Evidências de Causalidade <i>Orlando Carneiro de Matos</i>	Set/2002
50	Macroeconomic Coordination and Inflation Targeting in a Two-Country Model Eui Jung Chang, Marcelo Kfoury Muinhos and Joanílio Rodolpho Teixeira	Sep/2002
51	Credit Channel with Sovereign Credit Risk: an Empirical Test Victorio Yi Tson Chu	Sep/2002
52	Generalized Hyperbolic Distributions and Brazilian Data José Fajardo and Aquiles Farias	Sep/2002
53	Inflation Targeting in Brazil: Lessons and Challenges André Minella, Paulo Springer de Freitas, Ilan Goldfajn and Marcelo Kfoury Muinhos	Nov/2002
54	Stock Returns and Volatility Benjamin Miranda Tabak and Solange Maria Guerra	Nov/2002
55	Componentes de Curto e Longo Prazo das Taxas de Juros no Brasil Carlos Hamilton Vasconcelos Araújo e Osmani Teixeira de Carvalho de Guillén	Nov/2002
56	Causality and Cointegration in Stock Markets: the Case of Latin America Benjamin Miranda Tabak and Eduardo José Araújo Lima	Dec/2002
57	As Leis de Falência: uma Abordagem Econômica Aloisio Araujo	Dez/2002
58	The Random Walk Hypothesis and the Behavior of Foreign Capital Portfolio Flows: the Brazilian Stock Market Case <i>Benjamin Miranda Tabak</i>	Dec/2002
59	Os Preços Administrados e a Inflação no Brasil Francisco Marcos R. Figueiredo e Thaís Porto Ferreira	Dez/2002
60	Delegated Portfolio Management Paulo Coutinho and Benjamin Miranda Tabak	Dec/2002

61	O Uso de Dados de Alta Freqüência na Estimação da Volatilidade e do Valor em Risco para o Ibovespa João Maurício de Souza Moreira e Eduardo Facó Lemgruber	Dez/2002
62	Taxa de Juros e Concentração Bancária no Brasil Eduardo Kiyoshi Tonooka e Sérgio Mikio Koyama	Fev/2003
63	Optimal Monetary Rules: the Case of Brazil Charles Lima de Almeida, Marco Aurélio Peres, Geraldo da Silva e Souza and Benjamin Miranda Tabak	Feb/2003
64	Medium-Size Macroeconomic Model for the Brazilian Economy Marcelo Kfoury Muinhos and Sergio Afonso Lago Alves	Feb/2003
65	On the Information Content of Oil Future Prices <i>Benjamin Miranda Tabak</i>	Feb/2003
66	A Taxa de Juros de Equilíbrio: uma Abordagem Múltipla Pedro Calhman de Miranda e Marcelo Kfoury Muinhos	Fev/2003
67	Avaliação de Métodos de Cálculo de Exigência de Capital para Risco de Mercado de Carteiras de Ações no Brasil Gustavo S. Araújo, João Maurício S. Moreira e Ricardo S. Maia Clemente	Fev/2003
68	Real Balances in the Utility Function: Evidence for Brazil Leonardo Soriano de Alencar and Márcio I. Nakane	Feb/2003
69	r-filters: a Hodrick-Prescott Filter Generalization Fabio Araújo, Marta Baltar Moreira Areosa and José Alvaro Rodrigues Neto	Feb/2003
70	Monetary Policy Surprises and the Brazilian Term Structure of Interest Rates <i>Benjamin Miranda Tabak</i>	Feb/2003
71	On Shadow-Prices of Banks in Real-Time Gross Settlement Systems <i>Rodrigo Penaloza</i>	Apr/2003
72	O Prêmio pela Maturidade na Estrutura a Termo das Taxas de Juros Brasileiras Ricardo Dias de Oliveira Brito, Angelo J. Mont'Alverne Duarte e Osmani Teixeira de C. Guillen	Maio/2003
73	Análise de Componentes Principais de Dados Funcionais – uma Aplicação às Estruturas a Termo de Taxas de Juros Getúlio Borges da Silveira e Octavio Bessada	Maio/2003
74	Aplicação do Modelo de Black, Derman & Toy à Precificação de Opções Sobre Títulos de Renda Fixa Octavio Manuel Bessada Lion, Carlos Alberto Nunes Cosenza e César das Neves	Maio/2003
75	Brazil's Financial System: Resilience to Shocks, no Currency Substitution, but Struggling to Promote Growth <i>Ilan Goldfajn, Katherine Hennings and Helio Mori</i>	Jun/2003

76	Inflation Targeting in Emerging Market Economies Arminio Fraga, Ilan Goldfajn and André Minella	Jun/2003
77	Inflation Targeting in Brazil: Constructing Credibility under Exchange Rate Volatility André Minella, Paulo Springer de Freitas, Ilan Goldfajn and Marcelo Kfoury Muinhos	Jul/2003
78	Contornando os Pressupostos de Black & Scholes: Aplicação do Modelo de Precificação de Opções de Duan no Mercado Brasileiro <i>Gustavo Silva Araújo, Claudio Henrique da Silveira Barbedo, Antonio</i> <i>Carlos Figueiredo, Eduardo Facó Lemgruber</i>	Out/2003
79	Inclusão do Decaimento Temporal na Metodologia Delta-Gama para o Cálculo do VaR de Carteiras Compradas em Opções no Brasil Claudio Henrique da Silveira Barbedo, Gustavo Silva Araújo, Eduardo Facó Lemgruber	Out/2003
80	Diferenças e Semelhanças entre Países da América Latina: uma Análise de <i>Markov Switching</i> para os Ciclos Econômicos de Brasil e Argentina Arnildo da Silva Correa	Out/2003
81	Bank Competition, Agency Costs and the Performance of the Monetary Policy <i>Leonardo Soriano de Alencar and Márcio I. Nakane</i>	Jan/2004
82	Carteiras de Opções: Avaliação de Metodologias de Exigência de Capital no Mercado Brasileiro Cláudio Henrique da Silveira Barbedo e Gustavo Silva Araújo	Mar/2004
83	Does Inflation Targeting Reduce Inflation? An Analysis for the OECD Industrial Countries <i>Thomas Y. Wu</i>	May/2004
84	Speculative Attacks on Debts and Optimum Currency Area: a Welfare Analysis Aloisio Araujo and Marcia Leon	May/2004
85	Risk Premia for Emerging Markets Bonds: Evidence from Brazilian Government Debt, 1996-2002 <i>André Soares Loureiro and Fernando de Holanda Barbosa</i>	May/2004
86	Identificação do Fator Estocástico de Descontos e Algumas Implicações sobre Testes de Modelos de Consumo Fabio Araujo e João Victor Issler	Maio/2004
87	Mercado de Crédito: uma Análise Econométrica dos Volumes de Crédito Total e Habitacional no Brasil Ana Carla Abrão Costa	Dez/2004
88	Ciclos Internacionais de Negócios: uma Análise de Mudança de Regime Markoviano para Brasil, Argentina e Estados Unidos Arnildo da Silva Correa e Ronald Otto Hillbrecht	Dez/2004
89	O Mercado de <i>Hedge</i> Cambial no Brasil: Reação das Instituições Financeiras a Intervenções do Banco Central Fernando N. de Oliveira	Dez/2004

90	Bank Privatization and Productivity: Evidence for Brazil Márcio I. Nakane and Daniela B. Weintraub	Dec/2004
91	Credit Risk Measurement and the Regulation of Bank Capital and Provision Requirements in Brazil – a Corporate Analysis <i>Ricardo Schechtman, Valéria Salomão Garcia, Sergio Mikio Koyama and</i> <i>Guilherme Cronemberger Parente</i>	Dec/2004
92	Steady-State Analysis of an Open Economy General Equilibrium Model for Brazil <i>Mirta Noemi Sataka Bugarin, Roberto de Goes Ellery Jr., Victor Gomes</i> <i>Silva, Marcelo Kfoury Muinhos</i>	Apr/2005
93	Avaliação de Modelos de Cálculo de Exigência de Capital para Risco Cambial Claudio H. da S. Barbedo, Gustavo S. Araújo, João Maurício S. Moreira e Ricardo S. Maia Clemente	Abr/2005
94	Simulação Histórica Filtrada: Incorporação da Volatilidade ao Modelo Histórico de Cálculo de Risco para Ativos Não-Lineares Claudio Henrique da Silveira Barbedo, Gustavo Silva Araújo e Eduardo Facó Lemgruber	Abr/2005
95	Comment on Market Discipline and Monetary Policy by Carl Walsh <i>Maurício S. Bugarin and Fábia A. de Carvalho</i>	Apr/2005
96	O que É Estratégia: uma Abordagem Multiparadigmática para a Disciplina Anthero de Moraes Meirelles	Ago/2005
97	Finance and the Business Cycle: a Kalman Filter Approach with Markov Switching Ryan A. Compton and Jose Ricardo da Costa e Silva	Aug/2005
98	Capital Flows Cycle: Stylized Facts and Empirical Evidences for Emerging Market Economies <i>Helio Mori e Marcelo Kfoury Muinhos</i>	Aug/2005
99	Adequação das Medidas de Valor em Risco na Formulação da Exigência de Capital para Estratégias de Opções no Mercado Brasileiro Gustavo Silva Araújo, Claudio Henrique da Silveira Barbedo,e Eduardo Facó Lemgruber	Set/2005
100	Targets and Inflation Dynamics Sergio A. L. Alves and Waldyr D. Areosa	Oct/2005
101	Comparing Equilibrium Real Interest Rates: Different Approaches to Measure Brazilian Rates <i>Marcelo Kfoury Muinhos and Márcio I. Nakane</i>	Mar/2006
102	Judicial Risk and Credit Market Performance: Micro Evidence from Brazilian Payroll Loans Ana Carla A. Costa and João M. P. de Mello	Apr/2006
103	The Effect of Adverse Supply Shocks on Monetary Policy and Output Maria da Glória D. S. Araújo, Mirta Bugarin, Marcelo Kfoury Muinhos and Jose Ricardo C. Silva	Apr/2006

104	Extração de Informação de Opções Cambiais no Brasil <i>Eui Jung Chang e Benjamin Miranda Tabak</i>	Abr/2006
105	Representing Roommate's Preferences with Symmetric Utilities José Alvaro Rodrigues Neto	Apr/2006
106	Testing Nonlinearities Between Brazilian Exchange Rates and Inflation Volatilities <i>Cristiane R. Albuquerque and Marcelo Portugal</i>	May/2006
107	Demand for Bank Services and Market Power in Brazilian Banking Márcio I. Nakane, Leonardo S. Alencar and Fabio Kanczuk	Jun/2006
108	O Efeito da Consignação em Folha nas Taxas de Juros dos Empréstimos Pessoais Eduardo A. S. Rodrigues, Victorio Chu, Leonardo S. Alencar e Tony Takeda	Jun/2006
109	The Recent Brazilian Disinflation Process and Costs <i>Alexandre A. Tombini and Sergio A. Lago Alves</i>	Jun/2006
110	Fatores de Risco e o Spread Bancário no Brasil Fernando G. Bignotto e Eduardo Augusto de Souza Rodrigues	Jul/2006
111	Avaliação de Modelos de Exigência de Capital para Risco de Mercado do Cupom Cambial Alan Cosme Rodrigues da Silva, João Maurício de Souza Moreira e Myrian Beatriz Eiras das Neves	Jul/2006
112	Interdependence and Contagion: an Analysis of Information Transmission in Latin America's Stock Markets <i>Angelo Marsiglia Fasolo</i>	Jul/2006
113	Investigação da Memória de Longo Prazo da Taxa de Câmbio no Brasil Sergio Rubens Stancato de Souza, Benjamin Miranda Tabak e Daniel O. Cajueiro	Ago/2006
114	The Inequality Channel of Monetary Transmission Marta Areosa and Waldyr Areosa	Aug/2006
115	Myopic Loss Aversion and House-Money Effect Overseas: an Experimental Approach José L. B. Fernandes, Juan Ignacio Peña and Benjamin M. Tabak	Sep/2006
116	Out-Of-The-Money Monte Carlo Simulation Option Pricing: the Join Use of Importance Sampling and Descriptive Sampling <i>Jaqueline Terra Moura Marins, Eduardo Saliby and Joséte Florencio dos</i> <i>Santos</i>	Sep/2006
117	An Analysis of Off-Site Supervision of Banks' Profitability, Risk and Capital Adequacy: a Portfolio Simulation Approach Applied to Brazilian Banks Theodore M. Barnhill, Marcos R. Souto and Benjamin M. Tabak	Sep/2006
118	Contagion, Bankruptcy and Social Welfare Analysis in a Financial Economy with Risk Regulation Constraint <i>Aloísio P. Araújo and José Valentim M. Vicente</i>	Oct/2006

119	A Central de Risco de Crédito no Brasil: uma Análise de Utilidade de Informação Ricardo Schechtman	Out/2006
120	Forecasting Interest Rates: an Application for Brazil <i>Eduardo J. A. Lima, Felipe Luduvice and Benjamin M. Tabak</i>	Oct/2006
121	The Role of Consumer's Risk Aversion on Price Rigidity Sergio A. Lago Alves and Mirta N. S. Bugarin	Nov/2006
122	Nonlinear Mechanisms of the Exchange Rate Pass-Through: a Phillips Curve Model With Threshold for Brazil Arnildo da Silva Correa and André Minella	Nov/2006
123	A Neoclassical Analysis of the Brazilian "Lost-Decades" Flávia Mourão Graminho	Nov/2006
124	The Dynamic Relations between Stock Prices and Exchange Rates: Evidence for Brazil <i>Benjamin M. Tabak</i>	Nov/2006
125	Herding Behavior by Equity Foreign Investors on Emerging Markets Barbara Alemanni and José Renato Haas Ornelas	Dec/2006
126	Risk Premium: Insights over the Threshold José L. B. Fernandes, Augusto Hasman and Juan Ignacio Peña	Dec/2006
127	Uma Investigação Baseada em Reamostragem sobre Requerimentos de Capital para Risco de Crédito no Brasil Ricardo Schechtman	Dec/2006
128	Term Structure Movements Implicit in Option Prices <i>Caio Ibsen R. Almeida and José Valentim M. Vicente</i>	Dec/2006
129	Brazil: Taming Inflation Expectations Afonso S. Bevilaqua, Mário Mesquita and André Minella	Jan/2007
130	The Role of Banks in the Brazilian Interbank Market: Does Bank Type Matter? <i>Daniel O. Cajueiro and Benjamin M. Tabak</i>	Jan/2007
131	Long-Range Dependence in Exchange Rates: the Case of the European Monetary System Sergio Rubens Stancato de Souza, Benjamin M. Tabak and Daniel O. Cajueiro	Mar/2007
132	Credit Risk Monte Carlo Simulation Using Simplified Creditmetrics' Model: the Joint Use of Importance Sampling and Descriptive Sampling <i>Jaqueline Terra Moura Marins and Eduardo Saliby</i>	Mar/2007
133	A New Proposal for Collection and Generation of Information on Financial Institutions' Risk: the Case of Derivatives <i>Gilneu F. A. Vivan and Benjamin M. Tabak</i>	Mar/2007
134	Amostragem Descritiva no Apreçamento de Opções Européias através de Simulação Monte Carlo: o Efeito da Dimensionalidade e da Probabilidade de Exercício no Ganho de Precisão Eduardo Saliby, Sergio Luiz Medeiros Proença de Gouvêa e Jaqueline Terra Moura Marins	Abr/2007

135	Evaluation of Default Risk for the Brazilian Banking Sector <i>Marcelo Y. Takami and Benjamin M. Tabak</i>	May/2007
136	Identifying Volatility Risk Premium from Fixed Income Asian Options Caio Ibsen R. Almeida and José Valentim M. Vicente	May/2007
137	Monetary Policy Design under Competing Models of Inflation Persistence Solange Gouvea e Abhijit Sen Gupta	May/2007
138	Forecasting Exchange Rate Density Using Parametric Models: the Case of Brazil <i>Marcos M. Abe, Eui J. Chang and Benjamin M. Tabak</i>	May/2007
139	Selection of Optimal Lag Length inCointegrated VAR Models with Weak Form of Common Cyclical Features Carlos Enrique Carrasco Gutiérrez, Reinaldo Castro Souza and Osmani Teixeira de Carvalho Guillén	Jun/2007
140	Inflation Targeting, Credibility and Confidence Crises Rafael Santos and Aloísio Araújo	Aug/2007
141	Forecasting Bonds Yields in the Brazilian Fixed income Market Jose Vicente and Benjamin M. Tabak	Aug/2007
142	Crises Análise da Coerência de Medidas de Risco no Mercado Brasileiro de Ações e Desenvolvimento de uma Metodologia Híbrida para o <i>Expected Shortfall</i> Alan Cosme Rodrigues da Silva, Eduardo Facó Lemgruber, José Alberto Rebello Baranowski e Renato da Silva Carvalho	Ago/2007
143	Price Rigidity in Brazil: Evidence from CPI Micro Data Solange Gouvea	Sep/2007
144	The Effect of Bid-Ask Prices on Brazilian Options Implied Volatility: a Case Study of Telemar Call Options <i>Claudio Henrique da Silveira Barbedo and Eduardo Facó Lemgruber</i>	Oct/2007
145	The Stability-Concentration Relationship in the Brazilian Banking System <i>Benjamin Miranda Tabak, Solange Maria Guerra, Eduardo José Araújo</i> <i>Lima and Eui Jung Chang</i>	Oct/2007
146	Movimentos da Estrutura a Termo e Critérios de Minimização do Erro de Previsão em um Modelo Paramétrico Exponencial Caio Almeida, Romeu Gomes, André Leite e José Vicente	Out/2007
147	Explaining Bank Failures in Brazil: Micro, Macro and Contagion Effects (1994-1998) Adriana Soares Sales and Maria Eduarda Tannuri-Pianto	Oct/2007
148	Um Modelo de Fatores Latentes com Variáveis Macroeconômicas para a Curva de Cupom Cambial <i>Felipe Pinheiro, Caio Almeida e José Vicente</i>	Out/2007
149	Joint Validation of Credit Rating PDs under Default Correlation Ricardo Schechtman	Oct/2007

150	A Probabilistic Approach for Assessing the Significance of Contextual Variables in Nonparametric Frontier Models: an Application for	Oct/2007
	Brazilian Banks	
	Roberta Blass Staub and Geraldo da Silva e Souza	
151	Building Confidence Intervals with Block Bootstraps for the Variance	Nov/2007
	Ratio Test of Predictability	
	Eduardo José Araújo Lima and Benjamin Miranda Tabak	